\documentclass[11pt]{article}
\usepackage{amsfonts}
\usepackage{amsthm}
\usepackage{amsmath}
\usepackage{latexsym}
\usepackage{verbatim}
\usepackage[pdftex]{graphicx}
\usepackage{url}
\usepackage[table]{xcolor}
\usepackage[all,arc]{xy}
\usepackage{color, colortbl}
\usepackage{mathrsfs}
\usepackage{hyperref}
\hypersetup{colorlinks=true}
\setlength{\oddsidemargin}{.15in}
\setlength{\evensidemargin}{.15in}
\setlength{\textwidth}{6.2in}
\setlength{\topmargin}{-0.5in}
\setlength{\textheight}{8.6in}


\input{preamble.tex}

\begin{document}

\homework{9: Nash Equilibria}{Due Tuesday, week 10, at the start of class}



\begin{enumerate}
\item Find several examples of real-life situations that can be modeled with a payoff matrix.  Try to find Nash equilibria for each of these situations.  Do people in real life roughly seem to follow these equilibria, or do they do something different?  
\end{enumerate}




\end{document}